Author's response to reviews

Title: Animal-based folk remedies sold in public markets in Crato and Juazeiro do Norte, Ceara, Brazil.

Authors:

Felipe S Ferreira (ferreira_fs@yahoo.com.br)
Samuel V Brito (samuelvieirab@yahoo.com.br)
Samuel C Ribeiro (artsamus@hotmail.com)
Antônio AF Saraiva (alamocariri@yahoo.com.br)
Waltécio O Almeida (walmeida@urca.br)
Rômulo RN Alves (romulo_nobrega@yahoo.com.br)

Version: 3 Date: 26 May 2009

Author's response to reviews: see over
Dear editor, thanks for the suggestions. All points were answered and follow the specifications point to point:

**Major revisions (We require the author to make these changes)**

Reference 9: please provide full and accurate details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city. **DONE**

Reference 19, In press - Has this article now been published? If so, please provide the citation details.

**NO PUBLISHED**

Reference 26, In press - Has this article now been published? If so, please provide the citation details.

**ALREADY PUBLISHED AN THE CORRECT REFERENCE WAS ADDED**

Reference 28: please provide full and accurate details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city. **DONE**

Reference 42: please provide full and accurate details for this reference. This means for an Article providing volume/issue/page numbers and for a book the publisher/city. **DONE**

Table 2 too long - This table currently runs over multiple pages. We would prefer tables to fit onto a single A4 portrait page. If this is not possible please can you resubmit the tables as additional files, remembering to then remove the tables from the manuscript file. Please remember to then renumber any remaining tables accordingly, and to alter the references to the tables within the text.

**THE TABLE WAS MOVED AND SAVED AS ADDITIONAL FILES**

Reference 38: please provide a functioning web link for this reference. **DONE**

References - References should be cited in the text using consecutive numbers in square brackets. **DONE**

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table. **DONE**

Thanks again by the suggestions. I hope that enough.

Sincerely yours,
Felipe Ferreira